Author's response to reviews

Title: The risk of elevated prolactin levels in pediatric patients exposed to antipsychotics for the treatment of schizophrenia and other schizophrenia spectrum disorders: protocol for a systematic review and meta-analysis

Authors:

    Shawn Eapen (seapen@uottawa.ca)
    Eric Druyts (edruyts@uottawa.ca)
    Ping Wu (wuping629@hotmail.com)
    Kristian Thorlund (thorluk@mcmaster.ca)
    Edward J Mills (edward.mills@uottawa.ca)

Version: 2 Date: 9 July 2014

Author's response to reviews:

Dear Dr Moher,

Enclosed is our revised manuscript for your consideration at the Systematic Reviews Journal.

Note that funding for this study was provided by an external source (Canadian Institutes of Health Research). The protocol has been previously peer-reviewed.

Kind regards

Eric